# Is There a Magnet Effect of Rule-Based Circuit Breakers in Times of High-Frequency Trading?

Benjamin Clapham

 $Goe the \ University \ Frankfurt, \ clapham@wiwi.uni-frankfurt.de$ 

July 30, 2018

#### Abstract

This paper studies whether rule-based circuit breakers in the form of shortlived volatility interruptions exhibit a magnet effect in times of high-frequency trading. Based on a sample of 3,271 volatility interruptions on two major European venues, we analyze whether trading aggressiveness, trading activity, and volatility accelerate close to volatility interruptions indicating a magnet effect. Although the duration of the interruptions is meaningful given today's high-frequent securities markets, we do not find any evidence for a magnet effect. Rather, our results show that trading aggressiveness, trading activity, and volatility gradually slow down towards the triggering threshold and that price changes even revert in case of downward-triggered interruptions. These findings hold both for different levels of high-frequency trading activity and for disclosed and undisclosed price limits triggering the circuit breaker.

**Keywords:** Circuit Breaker, Volatility Interruption, Magnet Effect, Gravitational Effect, High-Frequency Trading

JEL Classification: G14, G15, G18

### 1 Introduction

Exchanges and regulatory authorities worldwide rely on market safeguards known as circuit breakers to ensure integrity and stability of securities markets. Particularly in times of increased automation of financial markets and the large proportion of algorithmic and high-frequency trading, exchange operators and regulators consider circuit breakers as suitable mechanisms to protect securities markets from sudden and large price jumps (Deutsche Boerse Group, 2016; European Parliament and Council, 2014). Moreover, increased volatility in financial markets and various high volatility events such as the May 2010 Flash Crash or the Brexit referendum in June 2016, which led to market turmoil, have accelerated the debate on circuit breakers among regulators, practitioners, and academics alike.

According to an international survey among exchanges, most developed securities markets have circuit breakers in place (Gomber et al., 2017). Moreover, regulators in different jurisdictions have conducted initiatives regarding the implementation or the design of market safeguards. The U.S. Securities and Exchange Commission (SEC), for example, enforced the implementation of the Limit Up-Limit Down pilot plan in 2012 in response to the May 2010 Flash Crash (U.S. Securities and Exchange Commission, 2012). Also, the revised Markets in Financial Instruments Directive (MiFID II), which came into effect in January 2018, makes the implementation of circuit breakers mandatory for trading venues within the European Union (European Parliament and Council, 2014). Most European trading venues rely on volatility interruptions that trigger short-lived instrument-specific call auctions if there is a significant price movement in a financial instrument (Gomber et al., 2017).

Although exchange operators and regulators view circuit breakers as something positive given their broad dissemination in securities markets, the academic perception regarding the effectiveness of circuit breakers is mixed and researchers also discuss possible negative effects of these safeguards. Among them is the so-called "magnet effect" of the price limits which trigger circuit breakers (Subrahmanyam, 1994). The magnet effect hypothesis assumes that traders accelerate their trading intention when prices come closer to the limit which actually pushes prices even further towards the limit and ultimately triggers the circuit breaker. Empirical studies come to contradicting results whether a magnet effect of circuit breakers exists. While papers analyzing longer interruptions mostly find evidence for a magnet effect, studies focusing on shorter interruptions do not. However, most of these studies base their analysis on data sets covering observation periods where no or only few high-frequency trading is present. Yet, high-frequency trading accounts for a large proportion of today's securities trading. According to a study conducted by the European Securities and Markets Authority (2014), high-frequency traders account for 30% to 49% of all trades in European equities markets. Due to increased activity and relative importance of high-frequency traders in securities markets over the past decade, it is of interest to investigate whether circuit breakers in the form of shortlived volatility interruptions exhibit a magnet effect in presence of high-frequency traders. For this group of traders, also a short interruption lasting only a few minutes represents a major constraint on their trading strategies that build upon order updates and trades within milliseconds.

Moreover, the implementation of circuit breakers differs among trading venues. In particular, some venues disclose the price limits triggering the interruption whereas others do not to avoid the magnet effect and to prevent traders from gaming the thresholds (Gomber et al., 2017). Therefore, the non-disclosure of triggering thresholds might mitigate the magnetic characteristic of rule-based circuit breakers since traders are unaware of the exact price limits. Nevertheless, sophisticated traders with access to large amounts of historical data can easily reverse engineer the unknown triggering thresholds and are thus able to derive the price limits of the circuit breaker. Consequently, we additionally investigate whether the disclosure or non-disclosure of the price limits triggering circuit breakers influences the magnet effect of rule-based circuit breakers.

In order to evaluate a potential magnet effect of circuit breakers in times of highfrequency trading, we analyze the trading period right before volatility interruptions on two major European exchanges. The analysis of volatility interruption mechanisms on two different venues not only increases the robustness of our results but also allows us to analyze our second research question. While the volatility interruption mechanisms are very similar on both venues, they differ in one important parameter, which is the disclosure respectively non-disclosure of the triggering thresholds.

Based on highly granular order book and transaction data, we use a sample of 3,271 volatility interruptions that occurred between 2011 and 2015 to analyze whether this rule-based circuit breaker exhibits a magnet effect in presence of high-frequency traders. According to the magnet effect hypothesis, traders are expected to adjust their trading behavior not only by advancing their trades in time, but also by trading more aggressively in proximity to the limits if they possess a magnetic attraction. In particular, a magnet effect would cause an acceleration of price developments, thus leading to larger price changes close to the limits, which ultimately lead to the surpassing of the triggering thresholds. Thus, a potential magnet effect should be observable in the form of higher trading aggressiveness, trading activity, and volatility close to the price limits.

Our results do not provide evidence for a magnet effect of short-lived volatility interruptions even in times of high-frequency trading. Instead of an acceleration of trading activity, trading aggressiveness, and volatility towards the limit, we rather observe trading and price changes to slow down close to the price limits triggering the volatility interruption. This finding holds for interruptions where high-frequency trading activity is high as well as for volatility interruptions where it is low. Moreover, we do not find a difference between volatility interruptions where the actual limits are disclosed to market participants and those interruptions where the market operator does not disclose the stock-specific limits.

The remainder of the paper proceeds as follows: Section 2 presents related literature on circuit breakers in general and on the magnet effect of circuit breakers in particular. Section 3 provides information on the institutional background, the data set, and descriptive statistics. Section 4 outlines the methodological approach as well as the results of our empirical analysis. The results are discussed in Section 5. Section 6 presents the conclusion.

### 2 Related Literature

#### 2.1 Circuit Breakers in General

Trading venues and regulatory authorities rely on different types of circuit breakers, i.e., market safeguards, to ensure integrity and stability of securities markets. An overview of the terminologies and concepts related to circuit breakers is provided by Abad and Pascual (2013). In general, circuit breakers can be divided into trading halts (also described as circuit breakers in a narrower sense), price limits, and volatility interruptions. In this paper, we focus on volatility interruptions, which suspend continuous trading of individual instruments with short-lived unscheduled call auctions.

There exists a large research stream on circuit breakers providing theoretical rationale and empirical evidence for both positive and negative effects of circuit breakers on market quality. From a theoretical point of view, circuit breakers give traders time to reassess their trading strategies and inventories, thus leading to reduced volatility and orderly trading after the circuit breaker (cooling-off hypothesis, introduced by Ma et al., 1989). Moreover, circuit breakers are beneficial in times of algorithmic trading since they can prevent flash crashes caused by automated submissions of and reactions to disruptive or erroneous orders (Subrahmanyam, 2013).

Nevertheless, circuit breakers have also been criticized by researchers and practitioners. Opponents of circuit breakers put forward that these mechanisms interfere with trading and market liquidity because traders cannot buy and sell instruments as needed and market makers may have problems to manage their inventories (Lauterbach and Ben-Zion, 1993). In addition, trading and/or price constraints induced by circuit breakers delay the incorporation of new information into prices and thus impede the price discovery process (Fama, 1970; Lehmann, 1989). Furthermore, circuit breakers might lead to volatility spillover to other markets and to subsequent trading periods (Subrahmanyam, 1994).

From an empirical point of view, the results concerning the effectiveness of circuit breakers to improve market quality are mixed. While some papers show that circuit breakers indeed reduce volatility (Abad and Pascual, 2010; Lee and Kim, 1995), others do not find evidence for a reduction of volatility (Bildik and Gülay, 2006; Kim et al., 2008). Moreover, some studies even show that volatility increases after a circuit breaker (Christie et al., 2002; Lee et al., 1994) and spills over to subsequent trading periods (Kim and Rhee, 1997). Within this paper, we focus on one specific point of criticism, which is the the so-called magnet effect of the price limits triggering circuit breakers. In the following subsection, we explain and discuss the magnet effect of circuit breakers in detail.

#### 2.2 The Magnet Effect of Circuit Breakers

The magnet effect hypothesis assumes that circuit breakers may cause traders to sub-optimally advance their trades in time once they are concerned about a likely trading constraint (Subrahmanyam, 1994). As a consequence, higher trading aggressiveness close to the price limits increases returns and price volatility, which effectively triggers the circuit breaker. Based on an experimental setup, Ackert et al. (2001) support the model of Subrahmanyam (1994) and provide additional experimental evidence that market participants advance their trades in time by comparing markets with no interruptions and markets with temporary trading halts.

In contrast to other hypotheses concerning circuit breakers, the magnet effect, which is also described as the gravitational effect, refers exclusively to the ex-ante effects of a circuit breaker (Cho et al., 2003). Specifically, the magnet effect predicts a change in trading behavior caused by the price limits of rule-based circuit breakers and describes market participants' herding behavior when there is a possibility of a limit hit (Abad and Pascual, 2013). According to the magnet effect hypothesis, a possible restriction of trading results in more aggressive trading since particularly day traders advance their trades in time to ensure that they can close their positions before the circuit breaker. Especially if hitting a price limit results in a long lasting circuit breaker or a complete trading halt, the magnet effect is reinforced. This trading behavior close to the triggering thresholds leads to ex-ante higher trading volume, more volatility, higher returns, and therefore increases the probability of hitting the price limit. A precondition for this effect is that the rule-based triggers of circuit breakers are publicly known or at least to some extent predictable. Randomizing circuit breakers and keeping price limits undisclosed can mitigate the possibility of a self-fulfilling interruption due to the magnet effect (Subrahmanyam, 1997).

The magnet effect has also been analyzed empirically for different markets, circuit breaker mechanisms, and observation periods. Early empirical work mostly finds no support for a magnet effect of price limits. Arak and Cook (1997) investigate the mornings after large overnight price fluctuations and test whether a magnet or a calming effect is caused by price limits. The authors develop two empirical models to disentangle the effect of price limits from the effect of incoming news and find a price reversal after the morning opening. This price reversal is sensitive to the proximity of prices to limits and provides evidence for a slight calming instead of a magnet effect. Berkman and Steenbeek (1998) study two exchanges, the Osaka Securities Exchange, which has strict price limits in place, and the Singapore International Monetary Exchange with looser limit regimes. They do not find any evidence for a magnetic attraction of price limits or systematic price movements towards the limits. Hall and Kofman (2001) analyze agricultural futures traded at the Chicago Board of Trade and find evidence for a stabilizing effect of price limits instead of magnetic attractions towards the limits. Also, Huang et al. (2001) do not find support for the magnet effect of price limits on the Taiwan Stock Exchange but rather attribute strong price movements to the overreaction of noise traders to new information. Regarding volatility interruptions, Abad and Pascual (2007) find that the probability of hitting a limit is very low even in close proximity to the respective limit. Furthermore, there rather is a reversion in prices since traders become more patient close to the triggering thresholds of the circuit breaker.

The majority of more recent work primarily finds positive evidence for a magnet effect of price limits. Several studies analyzing price limits on the Taiwan Stock Exchange show that they exhibit a magnet effect (Cho et al., 2003; Hsieh et al., 2009; Wong et al., 2009). In particular, Cho et al. (2003) find that upper limits exhibit a significant acceleration when prices converge towards the limit while lower limits show only weak evidence. Using data of price limits on the Korean Stock Exchange, Yan Du et al. (2009) show that acceleration rates increase as the distance to the price limit decreases and find that the effect is stronger for stricter price limits and weaker for quasi-limit hits. Thereby, the authors confirm that the effect is caused by magnetic attraction and not by momentum effects. A magnet effect of price limits is also shown for the Egyptian Stock Exchange and the Kuala Lumpur Stock Exchange (Chan et al., 2005; Tooma, 2011). Goldstein and Kavajecz (2004) investigate trading behavior and strategies on the New York Stock Exchange during the market turbulence and the market-wide trading halts in October 1997. Their findings show that traders alter the timing of their trades resulting in an acceleration of activity towards the circuit breaker, thus providing evidence for a magnet effect. To the best knowledge of the author, all existing empirical work concerning a potential magnet effect of circuit breakers focuses on observation periods before 2005 (in most cases even before 2000), where no or substantially less high-frequency trading was present. Moreover, existing studies are either based on daily data or intraday aggregations of low granularity, which do not allow to investigate a possible magnet effect of circuit breakers in presence of high-frequency traders, who might react within milliseconds close to the limit hit. Therefore, our paper adds to this research gap by analyzing whether circuit breakers in the form of short-lived volatility interruptions exhibit a magnet effect in presence of high-frequency traders. Since this group of traders acts and reacts within milliseconds, also an interruption of continuous trading for only a few minutes might impede trading strategies of highfrequency traders severely enough leading to a magnet effect of circuit breakers if high-frequency traders are active.

### 3 Data

#### 3.1 Institutional Background

In order to investigate a potential magnet effect of circuit breakers in presence of high-frequency traders, we analyze short-lived volatility interruptions on Deutsche Boerse's trading platform Xetra and the Spanish stock exchange Bolsa de Madrid (BME). Both markets are order-driven electronic open limit order books. This setup is particularly suitable for our empirical study since it allows us to analyze a possible magnet effect in the presence of high-frequency traders for two different but very similar volatility interruption mechanisms, which increases the robustness of our results. Moreover, we can investigate our second research question whether the disclosure or non-disclosure of triggering thresholds influences a possible magnet effect of circuit breakers. Table 1 shows major design parameters of the volatility interruptions implemented on Xetra and BME.

As described in Table 1, the general setup of volatility interruptions on Xetra and BME is very similar. In both cases, continuous trading is interrupted by an unscheduled call auction once the potential next execution price meets or exceeds the static threshold, which is based on the last auction price, or the dynamic threshold, which is based on the last trade price. During the auction of the volatility interruption, which is equipped with a randomized end, indicative prices and volumes are displayed to market participants. Also, the auction length of the two mechanisms is comparable although the duration of the volatility interruption on Xetra is a bit shorter than the interruption on BME (two minutes compared to five minutes).

Design of Volatility Interruptions on Xetra and BME								
This table provides detailed information about the design of volatility interruptions								
(volas) on Xetra and BME.								
Xetra	BME							
2:00 min	5:00 min							
0:30 min	0:30 min							
yes	no							
indicative price/volume	indicative price/volume							
not disclosed	4-10% <sup>1</sup>							
not disclosed	1-8%							
last auction price	last auction price							
Dynamic threshold last trade price last trade price								
	Xetra         Xetra         2:00 min         0:30 min         yes         indicative price/volume         not disclosed         not disclosed         last auction price         last trade price							

Table 1: Design of volatility interruptions on Xetra and BME.

Yet, only the auction phase on Xetra might be extended in case the potential execution price of the auction lies outside of a defined range, which is wider than the dynamic price range. The main difference, however, which is most important for the analysis of a potential magnet effect of short-lived volatility interruptions, is the disclosure respectively non-disclosure of the price limits that trigger the interruption. While these limits are disclosed on BME, they are not publicly known for the mechanism implemented on Xetra (Bolsa de Madrid, 2018; Deutsche Boerse Group, 2015). However, the limits can be approximated based on sufficient historical data.

#### **3.2** Data Set and Descriptive Statistics

We use tick-by-tick order book and trade information collected from Thomson Reuters Tick History for DAX30 and IBEX35<sup>2</sup> stocks traded on Xetra and BME. As the magnet effect is an ex-ante effect, we only consider the data prior to the interruption. Specifically, we analyze a period of 15 minutes prior to each volatility interruption and additionally consider a shorter observation window of five minutes to provide further robustness of our results. We split the data into 30 30-second intervals for the 15-minute observation period and into 30 ten-second intervals for the five-minute observation period. All trade-related variables are computed for each interval while order book information are averaged across the whole 15-minute (five-minute) observation period for each volatility interruption.

<sup>&</sup>lt;sup>1</sup>For the stock of Bankia S.A., the static threshold was raised up to 30% from May to August 2013 due to extraordinary high volatility.

<sup>&</sup>lt;sup>2</sup>Two constituents of the IBEX35 index, i.e., the stocks of Abengoa S.A. and Aena S.M.E. S.A., are not included in the analysis due to data issues.

Our data set includes all volatility interruptions in DAX30 and IBEX35 stocks that occurred between January 2011 and September 2015. Volatility interruptions are identified via unscheduled call auctions with suitable duration. Table 2 reports the number of volatility interruptions that occurred in the observation period. Since we consider a time frame of up to 15 minutes before a volatility interruption and in order to exclude possible confounding effects due to scheduled auctions, we exclude those volatility interruptions which started or ended within 15 minutes around opening, intraday<sup>3</sup>, or closing auctions. Furthermore, we exclude volatility interruptions where the pre-period overlaps a post-period of 15 minutes of a previous volatility interruption to prevent a potential bias of our results. Moreover, we exclude volatility interruptions with data issues and no observed trade within the 15-minute pre-period. This procedure results in 3,271 volatility interruptions in total, thereof 2,337 volatility interruptions on Xetra and 934 volatility interruptions on BME.

Number of Observed and Considered Volatility Interruptions								
Number of volatility interruptions (volas) on each venue during our observation								
period from January 1st, 2011 to September 30th, 2015 and detailed informa-								
tion about the actual number of interruptions used for our empirical analysis.								
Xetra BME Full Sample								
Total number of volas	3,048	1,131	4,179					
- Start of vola close to opening auction	248	92	340					
- Volas close to intraday auction	108	n.a.	108					
- End of vola close to closing auction	110	0	110					
- Volas close to end of another vola	240	101	341					
- Excluded volas due to data issues	5	4	9					
Number of considered volas	Number of considered volas 2,337 934 3,271							
Percentage of the sample	71.4%	28.6%	100.0%					

Table 2: Number of observed and considered volatility interruptions.

In contrast to other types of circuit beakers such as market-wide trading halts, which are rather rare events, single-stock volatility interruptions are frequently triggered on both markets Xetra and BME. Therefore, our data set is well-suited to analyze the magnet effect of circuit breakers based on a large number of observations. Table 3 provides descriptive statistics concerning the volatility interruptions included in the analysis. The distribution of considered volatility interruptions over the observation period and over the trading day is provided in Figures A.1 and A.2 in the appendix.

On average, 1.89 volatility interruptions occur in DAX30 stocks on Xetra per trading day. For IBEX35 stocks, the number is slightly lower amounting to 0.76 volatility interruptions per trading day. For each DAX30 (IBEX35) stock, we observe on average 77.90 (28.30) volatility interruptions during our observation period. With

<sup>&</sup>lt;sup>3</sup>There are no intraday auctions on BME. Consequently, only volatility interruptions that occurred on Xetra are excluded due to being close to intraday auctions.

Descriptive	Statistics	of the	Volatility	Interruptions
Descriptive	Dualistics	or the	volatility	interi uptions

This table reports descriptive statistics of all considered volatility interruptions (volas). Mean, median, minimum, and maximum values are computed over all considered volatility interruptions and are not pre-aggregated for each stock.

		Xet	ra		BME			
	Mean	Median	Min	Max	Mean	Median	Min	Max
Volas per day	1.89	1.00	0.00	50.00	0.76	0.00	0.00	25.00
Volas per stock	77.90	55.50	27.00	281.00	28.30	28.00	2.00	75.00
Duration [seconds]	135	135	120	253	314	314	300	330
Upward volas		1,008 (	(43%)		454 (49%)			
Downward volas		1.329 (	57%)		480 (51%)			

Table 3: Descriptive statistics of the volatility interruptions.

281 volatility interruptions, Commerzbank AG is the stock with the largest number of interruptions while the stock of Ferrovial S.A. is only interrupted twice. A detailed overview of the analyzed index constituents, the number of volatility interruptions per stock, and the stock-specific price limits triggering volatility interruptions is provided in Table A.1 in the appendix. Volatility interruptions on Xetra last on average 135 seconds, which equals the minimum duration of 120 seconds plus the expected value of the 30-second random end. Due to a possible extension of volatility interruptions on Xetra, the interruptions in the sample last up to 253 seconds. However, we only observe five volatility interruptions that last longer than 150 seconds and thus are classified as extensions. The duration of interruptions on BME is between 300 and 330 seconds with a mean of 314 seconds. Concerning the market movement in which the volatility interruptions are triggered, the share of volatility interruptions on Xetra triggered by the lower price limit is with 57% slightly higher than the share of interruptions triggered by the upper price limit (43%). For BME, the number of volatility interruptions is almost evenly split between downward (51%)and upward (49%) triggered interruptions.

The major difference between the volatility interruption mechanisms on Xetra and BME is that the price limits triggering the interruptions on BME are publicly known whereas the respective thresholds on Xetra are not disclosed to market participants. Nevertheless, the stock-specific price limits can be reverse engineered based on comprehensive historic trading data. Specifically, we calculate the static price limits by determining the largest possible deviation from the last auction price, which serves as the reference price for the upper and lower limits, within the 15 minutes before the volatility interruption is triggered. This procedure provides us with a valid approximation of the actual triggering thresholds since we study highly liquid blue chip stocks. In most cases, the largest deviation from the reference price is observed for the price of the last trade prior to the interruption. The constituents of the DAX30 and the IBEX35 index are frequently traded and price changes regularly appear in small increments. Thus, the computed quasi limit is only a small increment away from the actual limit. In addition, we validate our procedure based on the data and the disclosed price limits for volatility interruptions on BME. The validation results show that we are able to correctly approximate 81% of the thresholds. Since the approximated thresholds triggering volatility interruptions on Xetra feature a much lower variation than the thresholds on BME, the percentage of correctly approximated thresholds should be even higher for Xetra.

In order to study possible differences regarding a magnet effect of circuit breakers in presence of high and low high-frequency trading activity, we rely on the orderto-trade ratio (OTR) as a commonly used measure for high-frequency trading since publicly available data feeds do not include a specific flag for high-frequency trading activity (Brogaard et al., 2015; Haferkorn, 2017; Malinova et al., 2016). Furthermore, regulatory authorities also rely on the OTR as a measure for high-frequency trading and have passed acts that enforce trading venues to charge fees for traders with excessive OTRs (Friederich and Payne, 2015; German High-Frequency Trading Act, 2013). We measure the level of high-frequency trading activity based on the OTR during our two different observation periods of 15 and five minutes prior to each volatility interruption in our sample as shown in Equation (1).

$$OTR_i = \frac{Orders_i}{Trades_i}.$$
(1)

 $Orders_i$  represents the number of orders submitted to Xetra respectively BME during the 15-minute (five-minute) period prior to volatility interruption *i* while  $Trades_i$  is the number of executed trades. Since our data set does not build on order messages but on high-frequent order book snapshots, we indirectly obtain the number of order submissions in each interval following the methodology proposed by He et al. (2015). By comparing the number of orders on each limit to the previous order book situation, we compute the total number of orders that have been submitted<sup>4</sup>. For Xetra, we have to correct our approximation of the number of orders for volatility interruptions that occurred from April 7th, 2014 onwards since the data provider Thomson Reuters changed the data feed for Xetra from a netted feed into

<sup>&</sup>lt;sup>4</sup>With this procedure, we cannot observe market orders since they are immediately executed without an additional order book update between submission and execution. However, market orders are used to a small percentage in general and especially rarely by high-frequency traders. According to Jarnecic and Snape (2014), only 4.63% of high-frequency traders' orders are market orders.

a more granular feed on April 7th, 2014 (CEF Ultra $+^{5}$ ). There is no such change for BME during our observation period. Since the OTR shows a few very small and very large outliers, which might result from special market situations right before some of the volatility interruptions, we apply a 95% winsorization to the OTR, i.e., we replace the extreme values at both ends of the distribution with the value of the 2.5th (97.5th) percentile.

Trading Activity and Market Quality prior to Volatility Interruptions								
This table provides descriptive statistics regarding average trading activity and market quality 15 and five								
minutes prior to volatility interruptions. Trading volume and order book depth measured by Depth(10) are								
reported in million euro. Standard deviation of returns and relative spreads are reported in basis points.								
		Xet	ra			BM	Е	
	Mean	Median	Min	Max	Mean	Median	Min	Max
15 Minutes prior to interruption								
Number of trades	541	416	4	4,573	322	171	2	$4,\!540$
Trading volume [mn]	9.40	6.48	0.02	146.02	3.22	0.88	0.00	92.30
OTR	8.37	7.37	2.78	20.87	7.82	4.48	1.24	43.80
Rel. spread [bps]	8.12	6.91	2.14	81.64	25.08	18.18	2.74	162.11
Depth(10) [mn]	0.42	0.32	0.00	3.08	0.08	0.03	0.00	0.97
Std. dev. of returns [bps]	3.76	3.21	0.99	36.75	11.00	7.21	1.18	77.98
5 Minutes prior to interruption								
Number of trades	228	169	0	1,513	134	71	0	2,402
Trading volume [mn]	4.17	2.60	0.00	44.95	1.34	0.36	0.00	64.49
OTR	7.64	6.52	2.28	21.40	7.60	4.10	0.75	48.63
Rel. spread [bps]	8.21	6.89	2.21	96.45	25.22	17.67	2.36	192.10
Depth(10) [mn]	0.41	0.31	0.00	3.95	0.08	0.03	0.00	1.23
Std. dev. of returns [bps]	3.70	3.14	0.00	39.88	10.81	6.32	0.00	96.56

Table 4: Trading activity and market quality prior to volatility interruptions.

Descriptive statistics regarding trading activity and market quality prior to volatility interruptions are provided in Table 4. All variables are reported separately for each market and separately for the 15- and five-minute observation period prior to the interruption. On average, there is more trading activity prior to volatility interruptions on Xetra than prior to interruptions on BME, which holds for both mean number of trades and mean trading volume. Concerning the number of trades, there are on average 541 (228) trades on Xetra and only 322 (134) on BME in the 15 (five) minutes before a volatility interruption is triggered. The mean trading volume prior to volatility interruptions on Xetra amounts to 9.40 (4.17) and on BME to 3.22 (1.34) million euro in the 15-minute (five-minute) observation period. The OTR is quite similar on both markets with 8.37 versus 7.82 in the 15-minute period

<sup>&</sup>lt;sup>5</sup>Specifically, we correct the number of orders by a factor of 2.12 for volatility interruptions from April 7th, 2014 onwards, which equals the mean change in the number of approximated orders across all DAX30 stocks from the trading day prior to the change in the data feed to April 7th, 2014, which is the day when the change became effective (median = 2.13).

and 7.64 versus 7.60 in the five-minute period. With respect to market quality indicators, liquidity on Xetra is on average higher than on BME shortly before volatility interruptions, which particularly holds for liquidity in terms of order book depth, while volatility is higher on BME.

#### **3.3** Descriptive Analysis of the Magnet Effect

Having described the general characteristics of the data set in the previous subsection, we will now turn to the analysis of the magnet effect of rule-based circuit breakers in times of high-frequency trading in more detail and provide first descriptive findings within this subsection. As already explained, we consider an observation period of 15 (five) minutes prior to each volatility interruption and aggregate all trade-related information into 30-second (ten-second) intervals. In order to investigate whether traders change their behavior close to the price limits triggering volatility interruptions, we rely on five different measures quantifying trading activity, trading aggressiveness, and volatility. Trading activity is quantified by the number of trades and the trading volume in euro in each time interval. Trading aggressiveness is measured via the cumulative log return and the average trade size in each interval. Higher returns indicate that market participants trade more aggressively in a sense that they post more aggressive orders leading to higher market impact and that they are willing to trade at higher (lower) prices in case of a potential upper (lower) limit hit. Volatility is determined based on the standard deviation of returns in each interval.

Figure 1 depicts the development of the above mentioned five variables within the 15 minutes prior to the volatility interruption based on 30-second intervals and averaged across all considered volatility interruptions separately for each market. All variables measuring trading activity, trading aggressiveness, and volatility show a very similar pattern. Specifically, we observe a moderate increase over the 15 minutes prior to the interruption followed by a substantial jump in the last 30-second interval right before the volatility interruption is triggered. Regarding trading activity, a slightly higher increase in number of trades and trading volume is already visible from the fifth minute before the interruption onwards. We find very similar developments of trading activity, trading aggressiveness, and volatility also for the shorter fiveminute observation period, which is based on more granular ten-second intervals. These results are shown in Figure A.3 in the appendix. Consequently, the descriptive analysis building on the time-based distance to a volatility interruption shows some indication for a magnet effect due to the substantial jump in trading activity, trading aggressiveness, and volatility right before the interruption although no continuous acceleration over the whole observation window is observable.



Figure 1: Trading activity, trading aggressiveness, and volatility 15 minutes prior to volatility interruptions averaged across 30-second time intervals. Trading volume and trade size are reported in 1,000 euro. Interval returns and the standard deviation of returns are shown in bps.

However, the time-based distance to the volatility interruption is nothing traders can observe ex-ante since the exact timing of the volatility interruption is unknown and dependent on an incoming order leading to a potential execution price that meets or exceeds the limit. Therefore, we repeat the descriptive analysis by computing the relative distance (measured in bps) of each 30- respectively ten-second time interval to the price limit. We then average across all time intervals that belong to a certain distance interval in steps of 25 bps. The results of this analysis for the 15minute observation period are shown in Figure 2. Based on the distance to the price limit, no indications for a magnet effect can be observed. Rather, trading activity, cumulative interval returns, and volatility seem to decline the closer trade prices get to the limit triggering the volatility interruption. The average trade size remains on a constant level and also shows no indication for a magnetic attraction of the price limits triggering volatility interruptions. Again, the results remain robust for the five-minute observation period, which is depicted in Figure A.4 in the appendix.



Figure 2: Trading activity, trading aggressiveness, and volatility 15 minutes prior to volatility interruptions aggregated by the distance of each 30-second interval to the limits. Trading volume and trade size are reported in 1,000 euro. Interval returns and the standard deviation of returns are shown in bps.

### 4 Empirical Analysis

#### 4.1 Research Approach

In a first step, we calculate the relative distance to the upper (lower) price limit for each of the 30- respectively ten-second intervals in case of upward (downward) triggered volatility interruptions. Specifically, the relative distance to the upper price limit is defined as  $D_t^{UL} = (UL_t - P_t)/UL_t$ , where  $P_t$  is the first trade price<sup>6</sup> in time interval t and  $UL_t$  is the upper price limit. Similarly, the relative distance to the lower price limit is defined as  $D_t^{LL} = (P_t - LL_t)/LL_t$  with  $LL_t$  being the lower price limit.

For our empirical analysis, we rely on the method proposed by Abad and Pascual (2007) to analyze a possible magnet effect of volatility interruptions. We examine four value ranges for  $D_t^{UL}$  and  $D_t^{LL}$  in order to investigate whether trading activity, trading aggressiveness, and volatility accelerate towards the limit. Specifically, we consider the following ranges of distance to the limit (in bps):  $I_3 = [200,150), I_2 = [150,100), I_1 = [100,50), I_0 = [50,0]$  and run the regression model, which is shown in Equation (2), for each of our dependent variables measuring trading activity, trading aggressiveness, and volatility in a given time interval t of volatility interruption i.

$$Y_{i,t} = \alpha_0 + \alpha_1 \cdot Y_{i,t-1} + \sum_{k=n}^n \beta_k \cdot Controls_i + \sum_{d=0}^3 (\gamma_{i,d}^U \cdot U_{t-1}^d + \gamma_{i,d}^L \cdot L_{t-1}^d) + \varepsilon_{i,t}$$
(2)

As dependent variable  $Y_{i,t}$ , we use the cumulative return  $(R_t)$  as well as the mean trade size  $(S_t)$  in each 30- respectively ten-second interval to draw conclusions concerning a possible acceleration of trading aggressiveness. The trading volume in euro  $(V_t)$  and the number of trades  $(T_t)$  in each time interval are used to measure trading activity. Volatility as the last dependent variable is calculated as the standard deviation of returns  $(SD_t)$  in each interval. Since the dependent variables are determined simultaneously in time interval t, the error terms  $\varepsilon_{i,t}$  are expected to be contemporaneously correlated<sup>7</sup>. In order to account for this contemporaneous crossequation error correlation, we estimate Equation (2) for all dependent variables  $Y_{i,t}$ by a set of seemingly unrelated regression equations (SURE, Zellner, 1962) using the feasible generalized least squares (FGLS) procedure.

 $<sup>^{6}</sup>$ For robustness, we also use the last trade price in each time interval t to determine the relative distance to the upper respectively lower limit of a given 30- respectively ten-second interval and obtain very similar results. The results are available upon request.

<sup>&</sup>lt;sup>7</sup>We confirm the contemporaneous correlation of the error terms with Breusch-Pagan tests (Breusch and Pagan, 1979) in Section 4.2.

The variable of interest in this regression model is the dummy variable  $U_{t-1}^d$   $(L_{t-1}^d)$  that equals one when the first trade price in the previous time interval is within range  $D_d$  before an upper (lower) limit hit. The coefficients  $\gamma_{i,d}^U$  and  $\gamma_{i,d}^L$  thus allow us to draw conclusions regarding the existence of a magnet effect of the price limits triggering volatility interruptions. If a magnet effect is present and trading activity, trading aggressiveness, and volatility accelerate towards the limit, we expect  $\gamma_{i,3}^U < \gamma_{i,1}^U < \gamma_{i,0}^U$  for upper limit hits. For lower limit hits, we expect  $\gamma_{i,3}^L > \gamma_{i,2}^L > \gamma_{i,1}^L > \gamma_{i,0}^L$  regarding cumulative returns (since they become more negative if a magnet effect is present in case of downward triggered volatility interruptions) and  $\gamma_{i,3}^L < \gamma_{i,2}^L < \gamma_{i,1}^L < \gamma_{i,0}^L$  for the other four measures analyzed in this study.

In order to control for different levels of liquidity prior to volatility interruptions, we include the average relative spread, order book depth and order imbalance for the period prior to each volatility interruption as control variables.<sup>8</sup> Moreover, we control for the cumulative trading volume until the respective time interval of each interruption to account for systematic differences between volatility interruptions triggered in a very active or inactive trading environment. We run the regression model for both the 15-minute and the five-minute observation period prior to the interruption. Moreover, we repeat the regression for volatility interruptions triggered in case of high and low high-frequency trading activity and for interruptions with disclosed and undisclosed thresholds.

#### 4.2 Results

Table 5 reports the estimated  $\gamma_{i,d}^U$  and  $\gamma_{i,d}^L$  coefficients for the full sample of 3,271 volatility interruptions based on the 15-minute observation period. We do not find support for a magnet effect of price limits triggering rule-based circuit breakers in the form of volatility interruptions even in times of relevant high-frequency trading activity. This holds for both upward and downward triggered volatility interruptions. Instead of speeding up as the limit approaches, price changes slow down progressively the closer the interval is to the limit in case of upper limit hits, i.e.,  $\gamma_{i,3}^U > \gamma_{i,2}^U > \gamma_{i,1}^U > \gamma_{i,0}^U > 0$ . For lower limit hits, returns not only move towards zero but our results indicate that prices even revert within the closest interval to the price limit, i.e.,  $\gamma_{i,3}^L < \gamma_{i,1}^L < 0 < \gamma_{i,0}^L$ . These findings are in line with those of Abad and Pascual (2007).

Besides cumulative returns of the 30-second intervals, also mean trade size, number of trades, trading volume, and volatility progressively decrease the closer the

<sup>&</sup>lt;sup>8</sup>Order book depth is measured by Depth(X) with X = 10 bps around the midpoint as proposed by Degryse et al. (2015). Referring to Chordia et al. (2002), we calculate order imbalance as  $\frac{|Depth(10)_{Ask} - Depth(10)_{Bid}|}{Depth(10)}.$ 

respective interval is to the limit. Moreover, the estimated coefficients indicate that trade sizes, trading activity, and volatility are smaller respectively lower for time intervals within the price ranges up to 200 bps than for intervals which are farther away from the triggering threshold. These observations hold for both upper and lower limit hits, i.e.,  $0 > \gamma_{i,3}^U > \gamma_{i,2}^U > \gamma_{i,1}^U > \gamma_{i,0}^U$  and  $0 > \gamma_{i,3}^L > \gamma_{i,2}^L > \gamma_{i,1}^L$ . The results of the more granular ten-second intervals based on the shorter observation period of five minutes prior to the volatility interruption are almost identical and reported in Table A.2 in the appendix.

Trading Behavior Close to Volatility Interruptions (15 Minutes)										
This table reports the results of the regression model described in Equation (2) for the full										
sample of 3,271 volatility interruptions. The dummy variables $U_d$ ( $L_d$ ) indicate how far a										
30-second interval is away from the price limit in case of an upper (lower) limit hit. The										
ranges of	distance to the	limit (in bps) are $I_3$	$_{3} = [200, 150), h$	$I_2 = [150, 100), I_2$	$I_1 = [100, 50), I_0 =$					
[50,0]. Co	ontrols for liquid	lity, order imbalan	ce, and cumula	tive trading vol	lume are included.					
We provi	We provide t statistics in parentheses. Note: * $p < 0.05$ , ** $p < 0.01$ , *** $p < 0.001$ .									
	Trading A	ggressiveness	Trading	Activity	Volatility					
	Return $(R_t)$	Trade Size $(S_t)$	Trades $(T_t)$	Volume $(V_t)$	Volatility $(SD_t)$					
$U_3$	6.379***	-0.509***	-3.126***	$-56.51^{***}$	-0.457***					
	(19.86)	(-3.75)	(-11.01)	(-9.36)	(-5.99)					
$U_2$	5.651***	-0.596***	-4.066***	-77.93***	-0.529***					
	(20.50)	(-5.12)	(-16.67)	(-15.04)	(-8.07)					
$U_1$	4.692***	-0.967***	-5.126***	-93.59***	-0.644***					
-	(18.75)	(-9.14)	(-23.11)	(-19.86)	(-10.83)					
$U_0$	3.098***	-1.498***	-6.406***	-110.4***	-1.375***					
- 0	(11.91)	(-13.59)	(-27.71)	(-22.50)	(-22.14)					
$L_3$	-2.232***	-0.425***	-2.426***	$-52.41^{***}$	-0.334***					
0	(-7.73)	(-3.48)	(-9.49)	(-9.64)	(-4.86)					
La	-1.554***	-0.825***	-3.668***	-77.24***	-0.493***					
-	(-5.96)	(-7.47)	(-15.86)	(-15.71)	(-7.93)					
$L_1$	-0.709**	-1.132***	-4.660***	-95.36***	-0.599***					
-	(-2.80)	(-10.56)	(-20.77)	(-19.99)	(-9.94)					
$L_0$	$1.371^{***}$	-1.902***	-5.858***	-113.0***	-1.223***					
0	(4.81)	(-15.73)	(-23.16)	(-21.02)	(-18.00)					
Adj. $R^2$	0.019	0.366	0.434	0.451	0.219					
N			94,859							
Breusch-l	Pagan test: The	null of independer	nce is rejected a	at the 1% level	(p = 0.000).					

Table 5: Trading behavior close to volatility interruptions based on the full sample and the 15-minute observation period.

To shed further light on our primary research question of whether rule-based circuit breakers exhibit a magnet effect in times of high-frequency trading, we divide our sample into volatility interruptions that were triggered when high-frequency trading activity was high and those interruptions that were triggered when high-frequency trading activity was low. Therefore, we rely on the OTR as a measure of highfrequency trading activity and divide our full sample of 3,271 interruptions by the median OTR. We then run our regression model described in Equation (2) separately for both sub-samples. Table 6 reports the estimated coefficients  $\gamma_{i,d}^U$  and  $\gamma_{i,d}^L$ Again, we do not find any evidence for a magnet effect of rule-based circuit breakers in the form of short-lived volatility interruptions. Specifically, the results show that the slowing down respectively reversion of price changes and the decrease in trading aggressiveness, trading activity, and volatility towards the limits hold for interruptions triggered in phases both with a lot of high-frequency trading activity and less high-frequency trading activity. The implications of the results are identical to those of the full sample shown in Table 5. For robustness, similar results are obtained using the more granular data based on the five-minute observation period as well as splitting the data set for each market separately into events with high and low OTR. These results are provided in Tables A.3 - A.5 in the appendix.

Results Subdivided by High and Low High-Frequency Trading Activity (15 Minutes) This table reports the results of the regression model described in Equation (2) for sub-samples with high and low high-frequency trading activity. The dummy variables  $U_d$  ( $L_d$ ) indicate how far a 30-second interval is away from the price limit in case of an upper (lower) limit hit. The ranges of distance to the limit (in bps) are  $I_3 = [200,150)$ ,  $I_2 = [150,100)$ ,  $I_1 = [100,50)$ ,  $I_0 =$ [50,0]. Controls for liquidity, order imbalance, and cumulative trading volume are included. Note: \* p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001.

		Trading Aggressiveness		Trading	Activity	Volatility
HFT	Coef.	Return $(R_t)$	Trade Size $(S_t)$	Trades $(T_t)$	Volume $(V_t)$	Volatility $(SD_t)$
High	$U_3$	5.601***	-0.647***	-1.975***	-34.18***	-0.487***
	$U_2$	$5.085^{***}$	$-0.701^{***}$	$-2.478^{***}$	-46.26***	$-0.715^{***}$
	$U_1$	$4.485^{***}$	-1.005***	-3.390***	$-62.59^{***}$	$-0.771^{***}$
	$U_0$	$3.012^{***}$	$-1.464^{***}$	-4.990***	-84.43***	$-1.454^{***}$
	$L_3$	$-2.054^{***}$	0.034	$-1.465^{***}$	-21.88***	-0.366***
	$L_2$	$-1.466^{***}$	-0.809***	$-2.192^{***}$	-43.70***	$-0.588^{***}$
	$L_1$	-0.709**	$-1.074^{***}$	-3.536***	$-64.76^{***}$	$-0.762^{***}$
	$L_0$	$1.345^{***}$	$-1.914^{***}$	$-4.665^{***}$	-85.09***	-1.230***
Low	$U_3$	7.257***	-0.262	-3.392***	-64.02***	-0.393**
	$U_2$	$6.375^{***}$	-0.266	-4.709***	-93.33***	-0.260*
	$U_1$	$5.035^{***}$	-0.665***	$-6.155^{***}$	$-111.2^{***}$	-0.489***
	$U_0$	3.300***	$-1.224^{***}$	-7.303***	$-124.8^{***}$	$-1.269^{***}$
	$L_3$	-2.449***	$-0.749^{***}$	$-2.458^{***}$	$-67.26^{***}$	-0.179
	$L_2$	$-1.697^{***}$	-0.659***	$-4.214^{***}$	$-97.11^{***}$	$-0.264^{*}$
	$L_1$	-0.685	-1.040***	-4.855***	-114.0***	$-0.291^{**}$
	$L_0$	1.466**	$-1.776^{***}$	$-6.425^{***}$	$-131.7^{***}$	$-1.146^{***}$
Breuse	h-Pagan	tests: The null	of independence is	s rejected at th	e 1% level ( $p =$	0.000).

Table 6: Trading behavior close to volatility interruptions subdivided by high and low high-frequency trading activity based on the 15-minute observation period.

In a last step, we analyze whether the disclosure or non-disclosure of the thresholds triggering volatility interruptions has an influence on the magnetic attraction of these price limits. Therefore, we again divide our sample into sub-samples. As already discussed, the major difference between the volatility interruption mechanisms implemented on Xetra and BME is that only BME publishes the price limits triggering volatility interruptions. Consequently, we run the regression model separately for both markets and compare the estimated coefficients  $\gamma_{i,d}^U$  and  $\gamma_{i,d}^L$ . The results are reported in Table 7. Again, our results show that rule-based circuit breakers in the form of short lived volatility interruptions do not exhibit a magnet effect. In particular, the evidence against a magnet effect is independent of whether the price limits triggering the interruption are disclosed to market participants or not. Although a magnet effect of rule-based circuit breakers is theoretically more pronounced when the triggering thresholds are publicly known (Subrahmanyam, 1997), our results also indicate a slowing down of trading activity and price changes in case of interruptions with disclosed price limits. We obtain similar results using the shorter observation period of five minutes as shown in Table A.6 in the appendix.

Results Subdivided by Disclosed and Undisclosed Price Limits (15 Minutes)									
This table reports the results of the regression model described in Equation (2) for sub-samples with									
disclosed and undisclosed price limits. The dummy variables $U_d$ ( $L_d$ ) indicate how far a 30-second									
interval is away from the price limit in case of an upper (lower) limit hit. The ranges of distance to the									
limit (in bps) are $I_3 = [200,150), I_2 = [150,100), I_1 = [100,50), I_0 = [50,0]$ . Controls for liquidity, order									
imbalance, and cumulative trading volume are included. Note: * $p < 0.05$ , ** $p < 0.01$ , *** $p < 0.001$ .									
Trading Aggressiveness Trading Activity Volatility									
Price Limits	Coef.	Return $(R_t)$	Trade Size $(S_t)$	Trades $(T_t)$	Volume $(V_t)$	Volatility $(SD_t)$			
Disclosed	$U_3$	6.999***	0.083	$-2.647^{***}$	-15.75**	-1.189***			
	$U_2$	5.801***	-0.043	-3.348***	-29.98***	$-1.154^{***}$			
	$U_1$	$4.590^{***}$	-0.259*	$-4.437^{***}$	$-42.87^{***}$	-1.312***			
	$U_0$	$2.555^{***}$	$-0.717^{***}$	$-5.304^{***}$	-49.00***	-2.464***			
	$L_3$	-3.267***	-0.110	$-1.418^{**}$	$-15.96^{**}$	$-0.754^{**}$			
	$L_2$	$-1.735^{*}$	-0.142	$-2.246^{***}$	$-24.04^{***}$	-1.160***			
	$L_1$	-1.116	-0.459***	-3.233***	-40.30***	-1.067***			
	$L_0$	$1.556^{*}$	-0.926***	-4.257***	-47.33***	-2.207***			
Undisclosed	$U_3$	5.948***	-0.619***	-3.548***	-73.07***	-0.182***			
	$U_2$	$5.384^{***}$	-0.570***	-4.636***	-97.76***	-0.297***			
	$U_1$	$4.541^{***}$	-0.909***	$-5.782^{***}$	-115.8***	-0.378***			
	$U_0$	$3.291^{***}$	-1.215***	$-7.464^{***}$	-146.0***	-0.715***			
	$L_3$	$-1.899^{***}$	-0.505**	-2.800***	-67.06***	-0.156***			
	$L_2$	$-1.516^{***}$	-0.926***	-4.225***	-97.26***	-0.258***			
	$L_1$	-0.638**	-1.104***	$-5.517^{***}$	-118.1***	-0.424***			
	$L_0$	$1.006^{***}$	-1.738***	-6.998***	$-151.7^{***}$	-0.622***			
Breusch-Paga	n tests:	The null of inde	pendence is rejecte	ed at the 1% le	vel $(p = 0.000).$				

Table 7: Trading behavior close to volatility interruptions subdivided by disclosed and undisclosed price limits based on the 15-minute observation period. In general, our results show that the magnet effect, an often claimed negative effect of circuit breakers that leads to an acceleration of trading aggressiveness, trading activity, and volatility prior to the interruption, does not exist for short-lived volatility interruptions. This finding also holds when high-frequency traders, for whom the short-lived unscheduled auction still represents a major constraint on their trading strategies, are highly active. Moreover, a magnet effect cannot be observed even if the thresholds triggering the volatility interruption are publicly disclosed.

### 5 Discussion

While several academic papers analyzing short-lived circuit breakers do not find evidence in support of a magnet effect, these studies focus on time periods where no or substantially less high-frequency trading was present. For high-frequency traders, however, who react to new information within milliseconds and who regularly offset their inventories, also a seemingly short interruption of continuous trading for only two to five minutes represents a significant constraint on their trading possibilities. Thus, the question arises whether short-lived circuit breakers in the form of volatility interruptions exhibit a magnet effect in times of high-frequency trading.

Based on our analysis, we do not find evidence for a magnet effect of short-lived volatility interruptions implemented on two major European exchanges even in times of high-frequency trading. Instead of an acceleration of trading activity, trading aggressiveness, and volatility towards the price limits triggering the interruption, we rather observe these measures to slow down close to the limits. The slowing down of price changes and trading activity holds both for the full sample as well as for sub-samples of events with high and low high-frequency trading activity before the volatility interruption. In particular, our results support the findings of Abad and Pascual (2007) although we focus on shorter time intervals (i.e., a more granular aggregation based on 30 respectively ten seconds instead of five minutes) and a recent data set that includes substantial amounts of high-frequency trading volume.

Moreover, our results suggest that volatility interruptions do not exhibit a magnet effect independent of whether the price limits triggering volatility interruptions are publicly known or not. This finding is in line with Clapham et al. (2017), who show that the effectiveness of volatility interruptions to reduce volatility is not affected by the disclosure or non-disclosure of the triggering thresholds. Since professional market participants with access to comprehensive historical market data or sufficient experience are able to approximate the thresholds even if they are not disclosed, the disclosure of the price limits provides a level playing field for all market participants. Our results show that the disclosure of the limits does not reduce the effectiveness of the market safeguard due to a magnet effect. In line with this result, the majority of trading venues already publish the thresholds of their circuit breaker mechanisms (Gomber et al., 2017).

Our results are also relevant for practitioners such as exchange operators and regulators. We provide evidence that short-lived volatility interruptions do not seem to exhibit a magnet effect, which would potentially weaken the positive effects of the safeguard, even in today's high-frequent markets. Thus, our analysis provides further arguments for exchanges and regulators to implement respectively enforce the implementation of circuit breakers. Since a magnet effect representing a potential downside of these mechanisms can also not be observed in times of high-frequency trading, circuit breakers qualify as meaningful mechanisms to prevent sudden unexpected price movements in highly automated securities markets.

There are also some limitations connected to our study. Since Deutsche Boerse does not disclose the stock-specific price limits that trigger volatility interruptions, we have to reverse engineer them. Nevertheless, our procedure to approximate the thresholds based on the trade price that is farthest away from the reference price prior to the interruption appears to provide us with a valid approximation of the actual price limits since we study highly liquid blue chip stocks. In most cases, this is the price of the last trade prior to the interruption. The stocks which are analyzed in this study are frequently traded and price changes regularly appear in small increments. Thus, the actual limit is only a small increment away from the approximated limit. Moreover, we validate our procedure based on data for stocks traded on BME, where the actual price limits of the volatility interruptions are disclosed. Furthermore, our data set does not contain a flag indicating highfrequency trading activity so that we have to rely on the OTR as a proxy for the amount of high frequency trading. Nevertheless, this measure is very common in research concerning high-frequency trading (Brogaard et al., 2015; Haferkorn, 2017; Malinova et al., 2016) and is also used in regulatory acts aimed at restricting highfrequency trading activity (Friederich and Payne, 2015).

Regarding future research opportunities, the analysis of a potential magnet effect of circuit breakers in times of high-frequency trading can be extended to different types of circuit breakers. Since our analysis focuses on volatility interruptions that initiate an unscheduled call auction showing at least some pre-trade transparency in the form of indicative prices and execution volumes, our results are not necessarily transferable to other types of circuit breakers. Thus, future research could analyze a potential magnet effect of market-wide or single-stock trading halts based on data sets with relevant volumes of high-frequency traders. Also, future research might consider the possibility of market participants to trade on alternative trading venues or over the counter during a circuit breaker on the venue under investigation.

### 6 Conclusion

To the best knowledge of the author, this is the first paper testing the magnet effect of rule-based circuit breakers in presence of high-frequency traders. Specifically, we investigate whether the stock-specific price limits of volatility interruptions implemented on two major European exchanges exhibit a magnet effect. Volatility interruptions represent a rule-based circuit breaker as suggested by Madhavan (1992), which switches continuous trading into an unscheduled auction to avoid large price jumps and to handle market phases of high uncertainty. Although these unscheduled call auctions last only two to five minutes, they represent a major constraint for market participants such as high-frequency traders that react within milliseconds during continuous trading according to their different strategies. Therefore, we investigate whether these short-lived, rule-based circuit breakers exhibit a magnet effect in times of high-frequency trading, which accounts for a substantial part of overall trading volume in today's securities markets. Moreover, we analyze whether a non-disclosure of the actual limits mitigates a potential magnet effect although market participants with access to comprehensive historical market data can reverse engineer and estimate the price limits.

Based on a recent sample of 3,271 volatility interruptions on the German stock exchange Deutsche Boerse and the Spanish stock exchange Bolsa de Madrid, we do not find evidence for a magnet effect of short-lived volatility interruptions even in times of substantial high-frequency trading activity. Instead of an acceleration of trading activity, trading aggressiveness, and volatility towards the limit, we observe trading and price changes rather to slow down close to the limits triggering volatility interruptions. This finding holds for interruptions where high-frequency trading activity is high as well as for volatility interruptions where it is low. Moreover, we do not find a difference between volatility interruptions where the actual limits are disclosed to market participants and those interruptions where the market operator does not disclose the stock-specific limits. Consequently, market operators could publish the actual limits to create a level playing field among highly professional market participants that are able to estimate the limits based on historical data and participants that cannot without weakening the effectiveness of the market safeguard. In general, our findings provide further evidence that even in today's high-frequent securities markets, volatility interruptions do not exhibit a magnet effect, which is a regularly discussed potential negative effect of rule-based circuit breakers. Thus, volatility interruptions might serve as suitable market safeguards to handle phases of market stress as suggested by Madhavan (1992).

### References

- Abad, David and Roberto Pascual (2007). "On the Magnet Effect of Price Limits." In: European Financial Management 13.5, pp. 833–852.
- (2010). "Switching to a Temporary Call Auction in Times of High Uncertainty." In: Journal of Financial Research 33.1, pp. 45–75.
- (2013). "Holding Back Volatility: Circuit Breakers, Price Limits, and Trading Halts." In: Market Microstructure in Emerging and Developed Markets. Ed. by H. Kent Baker and Halil Kiymaz. The Robert W. Kolb Series in Finance. Hoboken, New Jersey: John Wiley & Sons, Inc, pp. 303–324.
- Ackert, Lucy F., Bryan K. Church, and Narayanan Jayaraman (2001). "An Experimental Study of Circuit Breakers: The Effects of Mandated Market Closures and Temporary Halts on Market Behavior." In: *Journal of Financial Markets* 4.2, pp. 185–208.
- Arak, Marcelle and Richard E. Cook (1997). "Do Daily Price Limits Act as Magnets? The Case of Treasury Bond Futures." In: *Journal of Financial Services Research* 12.1, pp. 5–20.
- Berkman, Henk and Onno W. Steenbeek (1998). "The Influence of Daily Price Limits on Trading in Nikkei Futures." In: *Journal of Futures Markets* 18.3, pp. 265–279.
- Bildik, Recep and Güzhan Gülay (2006). "Are Price Limits Effective? Evidence from the Istanbul Stock Exchange." In: Journal of Financial Research 29.3, pp. 383– 403.
- Bolsa de Madrid (2018). Market Model: Equities, Rights and Latibex Market. URL: http://www.bolsamadrid.es/docs/SBolsas/docsSubidos/SIBE/marketmode l.pdf (visited on 05/09/2018).
- Breusch, Trevor S. and Adrian R. Pagan (1979). "A Simple Test for Heteroscedasticity and Random Coefficient Variation." In: *Econometrica* 47.5, pp. 1287–1294.
- Brogaard, Jonathan, Björn Hagströmer, Lars Nordén, and Ryan Riordan (2015). "Trading Fast and Slow: Colocation and Liquidity." In: *Review of Financial Studies* 28.12, pp. 3407–3443.
- Chan, Soon Huat, Kenneth A. Kim, and S. Ghon Rhee (2005). "Price Limit Performance: Evidence from Transactions Data and the Limit Order Book." In: *Journal of Empirical Finance* 12.2, pp. 269–290.
- Cho, David D., Jeffrey Russell, George C. Tiao, and Ruey Tsay (2003). "The Magnet Effect of Price Limits: Evidence from High-Frequency Data on Taiwan Stock Exchange." In: *Journal of Empirical Finance* 10.1-2, pp. 133–168.
- Chordia, Tarun, Richard Roll, and Avanidhar Subrahmanyam (2002). "Order Imbalance, Liquidity, and Market Returns." In: Journal of Financial Economics 65.1, pp. 111–130.

- Christie, William G., Shane A. Corwin, and Jeffrey H. Harris (2002). "Nasdaq Trading Halts: The Impact of Market Mechanisms on Prices, Trading Activity, and Execution Costs." In: *The Journal of Finance* 57.3, pp. 1443–1478.
- Clapham, Benjamin, Peter Gomber, Martin Haferkorn, and Sven Panz (2017). "Managing Excess Volatility: Design and Effectiveness of Circuit Breakers." In: Working Paper.
- Degryse, Hans, Frank de Jong, and Vincent van Kervel (2015). "The Impact of Dark Trading and Visible Fragmentation on Market Quality." In: *Review of Finance* 19.4, pp. 1587–1622.
- Deutsche Boerse Group (2015). Xetra Release 16.0: Market Model Equities. URL: http://www.deutsche-boerse-cash-market.com/blob/1193332/fc4798a 519e3e9c91eed90222840cf52/data/xetra-deutsche-boerse-cash-marketmarket-model-equities.pdf (visited on 05/09/2018).
- (2016). Protection in Times of Volatility. URL: http://www.deutsche-boersecash-market.com/dbcm-en/newsroom/focus-news/Martin-Reck-Interview -ETF-Magazin/2729032 (visited on 05/29/2018).
- European Parliament and Council (2014). Markets in Financial Instruments Directive 2014/65/EU: MiFID II.
- European Securities and Markets Authority (2014). High-Frequency Trading Activity in EU Equity Markets. URL: https://www.esma.europa.eu/sites/default/ files/library/2015/11/esma20141\_-\_hft\_activity\_in\_eu\_equity\_market s.pdf (visited on 05/07/2018).
- Fama, Eugene F. (1970). "Efficient Capital Markets: A Review of Theory and Empirical Work." In: The Journal of Finance 25.2, pp. 383–417.
- Friederich, Sylvain and Richard Payne (2015). "Order-to-Trade Ratios and Market Liquidity." In: Journal of Banking & Finance 50, pp. 214–223.
- German High-Frequency Trading Act (2013). Gesetz zur Vermeidung von Gefahren und Missbräuchen im Hochfrequenzhandel (Hochfrequenzhandelsgesetz). URL: h ttp://www.bgbl.de/xaver/bgbl/start.xav?startbk=Bundesanzeiger\_ BGBl&jumpTo=bgbl113s1162.pdf#\_\_bgbl\_\_%2F%2F\*%5B%40attr\_id%3D% 27bgbl113s1162.pdf%27%5D\_\_1480327147457 (visited on 05/23/2018).
- Goldstein, Michael A. and Kenneth A. Kavajecz (2004). "Trading Strategies during Circuit Breakers and Extreme Market Movements." In: Journal of Financial Markets 7.3, pp. 301–333.
- Gomber, Peter, Benjamin Clapham, Martin Haferkorn, Sven Panz, and Paul Jentsch (2017). "Ensuring Market Integrity and Stability: Circuit Breakers on International Trading Venues." In: *The Journal of Trading* 12.1, pp. 42–54.
- Haferkorn, Martin (2017). "High-Frequency Trading and its Role in Fragmented Markets." In: Journal of Information Technology 32.3, pp. 283–296.

- Hall, Anthony D. and Paul Kofman (2001). "Limits to Linear Price Behavior: Futures Prices Regulated by Limits." In: *Journal of Futures Markets* 21.5, pp. 463– 488.
- He, Peng William, Elvis Jarnecic, and Yubo Liu (2015). "The Determinants of Alternative Trading Venue Market Share: Global Evidence from the Introduction of Chi-X." In: *Journal of Financial Markets* 22, pp. 27–49.
- Hsieh, Ping-Hung, Yong H. Kim, and J. Jimmy Yang (2009). "The Magnet Effect of Price Limits: A Logit Approach." In: *Journal of Empirical Finance* 16.5, pp. 830– 837.
- Huang, Yen-Sheng, Tze-Wei Fu, and Mei-Chu Ke (2001). "Daily Price Limits and Stock Price Behavior: Evidence from the Taiwan Stock Exchange." In: International Review of Economics & Finance 10.3, pp. 263–288.
- Jarnecic, Elvis and Mark Snape (2014). "The Provision of Liquidity by High-Frequency Participants." In: *Financial Review* 49.2, pp. 371–394.
- Kim, Kenneth A. and S. Ghon Rhee (1997). "Price Limit Performance: Evidence from the Tokyo Stock Exchange." In: *The Journal of Finance* 52.2, pp. 885–901.
- Kim, Yong H., José Yagüe, and J. Jimmy Yang (2008). "Relative Performance of Trading Halts and Price Limits: Evidence from the Spanish Stock Exchange." In: International Review of Economics & Finance 17.2, pp. 197–215.
- Lauterbach, Beni and Uri Ben-Zion (1993). "Stock Market Crashes and the Performance of Circuit Breakers: Empirical Evidence." In: *The Journal of Finance* 48.5, pp. 1909–1925.
- Lee, Charles M.C., Mark J. Ready, and Paul J. Seguin (1994). "Volume, Volatility, and New York Stock Exchange Trading Halts." In: *The Journal of Finance* 49.1, pp. 183–214.
- Lee, Sang-Bin and Kwang-Jung Kim (1995). "The Effect of Price Limits on Stock Price Volatility: Empirical Evidence in Korea." In: Journal of Business Finance & Accounting 22.2, pp. 257–267.
- Lehmann, Bruce N. (1989). "Commentary: Volatility, Price Resolution, and the Effectiveness of Price Limits." In: Journal of Financial Services Research 3.2-3, pp. 205–209.
- Ma, Christopher K., Ramesh P. Rao, and R. Stephen Sears (1989). "Limit Moves and Price Resolution: The Case of the Treasury Bond Futures Market." In: *Journal* of Futures Markets 9.4, pp. 321–335.
- Madhavan, Ananth (1992). "Trading Mechanisms in Securities Markets." In: *The Journal of Finance* 47.2, pp. 607–641.
- Malinova, Katya, Andreas Park, and Ryan Riordan (2016). "Taxing High Frequency Market Making: Who Pays the Bill?" In: *Working Paper*.
- Subrahmanyam, Avanidhar (1994). "Circuit Breakers and Market Volatility: A Theoretical Perspective." In: *Journal of Finance* 49.1, pp. 237–254.

- Subrahmanyam, Avanidhar (1997). "The ex ante Effects of Trade Halting Rules on Informed Trading Strategies and Market Liquidity." In: *Review of Financial Economics* 6.1, pp. 1–14.
- (2013). "Algorithmic Trading, the Flash Crash, and Coordinated Circuit Breakers." In: *Borsa Istanbul Review* 13.3, pp. 4–9.
- Tooma, Eskandar A. (2011). "The Magnetic Attraction of Price Limits." In: *International Journal of Business* 16.1, pp. 35–50.
- U.S. Securities and Exchange Commission (2012). Investor Bulletin: Measures to Address Market Volatility. URL: https://www.sec.gov/oiea/investoralerts-bulletins/investor-alerts-circuitbreakersbulletinhtm.html (visited on 05/29/2018).
- Wong, Woon K., Matthew C. Chang, and Anthony H. Tu (2009). "Are Magnet Effects Caused by Uninformed Traders? Evidence from Taiwan Stock Exchange." In: *Pacific-Basin Finance Journal* 17.1, pp. 28–40.
- Yan Du, Daphne, Qianqiu Liu, and S. Ghon Rhee (2009). "An Analysis of the Magnet Effect under Price Limits." In: International Review of Finance 9.1-2, pp. 83–110.
- Zellner, Arnold (1962). "An Efficient Method of Estimating Seemingly Unrelated Regressions and Tests for Aggregation Bias." In: Journal of the American Statistical Association 57.298, pp. 348–368.

## Appendix



Figure A.1: Distribution of volatility interruptions over the observation period and major events leading to spikes in the number of interruptions per day.



Figure A.2: Distribution of volatility interruptions over the trading day.

(volas) per stoc	k during	Our obse Xetra	rvation	period, a	nd the co	prresponding pric	e limits t	BME	g volatili	ity interru	iptions.
		U	р	Dov	wn			U	р	Do	wn
Instrument	Volas	Mean	Max	Mean	Min	Instrument	Volas	Mean	Max	Mean	Min
ADSGn.DE	48	3%	4%	-3%	-4%	ABE.MC	11	4%	5%	-5%	-5%
ALVG.DE	54	4%	4%	-3%	-4%	ACS.MC	36	5%	7%	-5%	-7%
BASFn.DE	52	3%	4%	-3%	-4%	ACX.MC	30	5%	5%	-5%	-5%
BAYGn.DE	57	3%	5%	-3%	-5%	AMA.MC	5	6%	6%	-5%	-6%
BEIG.DE	29	3%	3%	-3%	-4%	ANA.MC	38	5%	7%	-5%	-7%
BMWG.DE	94	3%	4%	-3%	-4%	BBVA.MC	28	6%	6%	-6%	-6%
CBKG.DE	281	3%	4%	-3%	-6%	BKIA.MC	71	9%	20%	-9%	-30%
CONG.DE	65	4%	6%	-4%	-6%	BKT.MC	30	6%	7%	-6%	-7%
DAIGn.DE	90	3%	4%	-3%	-4%	CABK.MC	30	5%	6%	-5%	-6%
DB1Gn.DE	52	3%	4%	-3%	-4%	DIDA.MC	7	8%	8%	-6%	-7%
DBKGn.DE	149	3%	5%	-3%	-5%	ELE.MC	19	4%	5%	-5%	-5%
DPWGn.DE	38	3%	4%	-3%	-4%	ENAG.MC	7	n.a.	n.a.	-4%	-5%
DTEGn.DE	39	3%	4%	-3%	-4%	FCC.MC	75	5%	7%	-5%	-7%
EONGn.DE	86	3%	4%	-3%	-4%	FER.MC	2	n.a.	n.a.	-6%	-6%
FMEG.DE	32	3%	4%	-3%	-4%	GAM.MC	56	6%	8%	-6%	-8%
FREG.DE	41	3%	4%	-3%	-4%	GAS.MC	24	5%	5%	-4%	-5%
HEIG.DE	107	3%	4%	-3%	-4%	GRLS.MC	21	5%	5%	-5%	-5%
HNKG_p.DE	27	3%	4%	-3%	-4%	IBE.MC	22	5%	6%	-5%	-6%
IFXGn.DE	114	3%	4%	-3%	-5%	ICAG.MC	10	7%	7%	-7%	-7%
LHAG.DE	137	3%	4%	-3%	-4%	IDR.MC	33	6%	10%	-4%	-6%
LING.DE	37	3%	4%	-3%	-4%	ITX.MC	10	6%	6%	-5%	-6%
LXSG.DE	87	4%	6%	-4%	-6%	MAP.MC	40	5%	6%	-5%	-6%
MRCG.DE	36	3%	4%	-3%	-4%	MTS.MC	35	5%	6%	-5%	-6%
MUVGn.DE	32	3%	4%	-3%	-4%	OHL.MC	33	6%	6%	-6%	-6%
RWEG.DE	108	3%	4%	-3%	-4%	POP.MC	44	6%	8%	-6%	-7%
SAPG.DE	28	3%	4%	-3%	-4%	REE.MC	11	4%	5%	-5%	-5%
SDFGn.DE	139	3%	4%	-3%	-4%	REP.MC	29	5%	6%	-5%	-5%
SIEGn.DE	37	3%	4%	-3%	-4%	SABE.MC	38	6%	6%	-6%	-6%
TKAG.DE	128	3%	4%	-3%	-4%	SAN.MC	24	5%	6%	-6%	-6%
VOWG_p.DE	113	3%	4%	-3%	-4%	SCYR.MC	57	8%	8%	-7%	-8%
						TEF.MC	12	5%	10%	-5%	-5%
						TL5.MC	26	6%	7%	-6%	-7%
						TRE.MC	20	6%	7%	-6%	-7%
No. of volas	2,337						934				

Considered Index Constituents, Number of Volatility Interruptions, and Price Limits This table provides all considered DAX30 and IBEX35 constituents, the number of analyzed volatility interruptions

Table A.1: Considered index constituents, number of volatility interruptions, and price limits triggering volatility interruptions.



Figure A.3: Trading activity, trading aggressiveness, and volatility five minutes prior to volatility interruptions averaged across ten-second time intervals. Trading volume and trade size are reported in 1,000 euro. Interval returns and the standard deviation of returns are shown in bps.



Figure A.4: Trading activity, trading aggressiveness, and volatility five minutes prior to volatility interruptions aggregated by the distance of each ten-second interval to the limits. Trading volume and trade size are reported in 1,000 euro. Interval returns and the standard deviation of returns are shown in bps.

This table	This table reports the results of the regression model described in Equation (2) for the full								
sample of 3,271 volatility interruptions. The dummy variables $U_d$ ( $L_d$ ) indicate how far a									
ten-second interval is away from the price limit in case of an upper (lower) limit hit. The									
ranges of distance to the limit (in bps) are $I_3 = [200, 150), I_2 = [150, 100), I_1 = [100, 50), I_0 = $									
[50,0]. Controls for liquidity, order imbalance, and cumulative trading volume are included.									
We provide t statistics in parentheses. Note: * $p < 0.05$ , ** $p < 0.01$ , *** $p < 0.001$ .									
	Trading A	ggressiveness	Trading	Activity	Volatility				
	Return $(R_t)$	Trade Size $(S_t)$	Trades $(T_t)$	Volume $(V_t)$	Volatility $(SD_t)$				
$U_3$	$6.408^{***}$	-0.384	$-2.223^{***}$	$-48.48^{***}$	$-0.356^{***}$				
	(20.73)	(-1.61)	(-10.07)	(-7.41)	(-4.06)				
$U_2$	5.723***	-0.0635	-2.858***	-63.63***	-0.434***				
	(23.11)	(-0.33)	(-16.15)	(-12.14)	(-6.18)				
$U_1$	4.646***	-0.927***	-3.719***	-81.71***	-0.625***				
	(23.47)	(-6.08)	(-26.23)	(-19.48)	(-11.14)				
$U_0$	3.051***	-2.273***	-4.783***	-101.3***	-1.157***				
	(17.22)	(-16.63)	(-37.55)	(-26.94)	(-23.00)				
$L_3$	$-1.259^{***}$	-0.265	$-1.564^{***}$	-53.18***	-0.109				
	(-4.85)	(-1.33)	(-8.43)	(-9.66)	(-1.48)				
$L_2$	-0.381	-0.501**	$-2.451^{***}$	-64.70***	-0.371***				
	(-1.81)	(-3.08)	(-16.25)	(-14.47)	(-6.21)				
$L_1$	$0.406^{*}$	-1.221***	-3.276***	-82.61***	-0.550***				
	(2.14)	(-8.35)	(-24.09)	(-20.52)	(-10.22)				
$L_0$	1.646***	-2.483***	-4.484***	-106.7***	-1.079***				
	(8.93)	(-17.43)	(-33.84)	(-27.24)	(-20.60)				
Adj. $R^2$	0.025	0.237	0.305	0.234	0.139				
N			93,699						
Breusch-H	Pagan test: The	null of independer	nce is rejected a	at the 1% level	(p = 0.000).				

Trading Behavior Close to Volatility Interruptions (5 Minutes)

Table A.2: Trading behavior close to volatility interruptions based on the full sample and the five-minute observation period.

Results Subdivided by High and Low High-Frequency Trading Activity (5 Minutes) This table reports the results of the regression model described in Equation (2) for sub-samples with high and low high-frequency trading activity. The dummy variables  $U_d$  ( $L_d$ ) indicate how far a ten-second interval is away from the price limit in case of an upper (lower) limit hit. The ranges of distance to the limit (in bps) are  $I_3 = [200,150)$ ,  $I_2 = [150,100)$ ,  $I_1 = [100,50)$ ,  $I_0 = [50,0]$ . Controls for liquidity, order imbalance, and cumulative trading volume are included. Note: \* p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001.

		Trading A	Trading Aggressiveness		Activity	Volatility
HFT	Coef.	Return $(R_t)$	Trade Size $(S_t)$	Trades $(T_t)$	Volume $(V_t)$	Volatility $(SD_t)$
High	$U_3$	5.276***	-0.586	-1.350***	-33.65***	-0.359***
	$U_2$	$5.023^{***}$	-0.544	$-1.591^{***}$	-41.48***	$-0.550^{***}$
	$U_1$	$4.026^{***}$	-1.210***	-2.308***	-58.54***	-0.680***
	$U_0$	$2.577^{***}$	$-2.253^{***}$	$-3.241^{***}$	$-73.91^{***}$	-1.180***
	$L_3$	$-1.851^{***}$	-0.347	$-0.877^{***}$	$-45.77^{***}$	-0.099
	$L_2$	-0.706**	-0.905***	$-1.516^{***}$	$-52.11^{***}$	-0.460***
	$L_1$	0.195	$-1.397^{***}$	-2.093***	$-62.58^{***}$	-0.645***
	$L_0$	$1.264^{***}$	-2.523***	-3.041***	-81.02***	-1.123***
Low	$U_3$	7.041***	-0.117	-2.289***	-43.83***	-0.338*
	$U_2$	$6.079^{***}$	0.398	$-3.112^{***}$	$-64.50^{***}$	$-0.257^{*}$
	$U_1$	$5.088^{***}$	$-0.489^{*}$	-4.004***	-83.73***	-0.498***
	$U_0$	$3.369^{***}$	$-1.934^{***}$	$-5.150^{***}$	$-103.1^{***}$	-1.038***
	$L_3$	$-0.997^{*}$	-0.209	$-1.346^{***}$	-50.62***	-0.006
	$L_2$	-0.339	-0.097	$-2.276^{***}$	$-62.67^{***}$	-0.168
	$L_1$	0.339	-0.920***	$-3.194^{***}$	-86.64***	$-0.341^{***}$
	$L_0$	$1.799^{***}$	$-2.159^{***}$	$-4.663^{***}$	-111.8***	-0.923***
Breuse	h-Pagan	tests: The null	of independence is	s rejected at th	e 1% level ( $p =$	0.000).

Table A.3: Trading behavior close to volatility interruptions subdivided by high and low high-frequency trading activity based on the five-minute observation period.

Results Subdivided by Market and High and Low HFT Activity (15 Minutes) This table reports the results of the regression model described in Equation (2) for sub-samples with high and low high-frequency trading activity separately for each market. The dummy variables  $U_d$  $(L_d)$  indicate how far a 30-second interval is away from the price limit in case of an upper (lower) limit hit. The ranges of distance to the limit (in bps) are  $I_3 = [200,150)$ ,  $I_2 = [150,100)$ ,  $I_1 = [100,50)$ ,  $I_0 = [50,0]$ . Controls for liquidity, order imbalance, and cumulative trading volume are included. Note: \* p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001.

		Trading A	ggressiveness	Trading	Activity	Volatility
HFT	Coef.	Return $(R_t)$	Trade Size $(S_t)$	Trades $(T_t)$	Volume $(V_t)$	Volatility $(SD_t)$
			Xe	,		
High	$U_3$	$5.239^{***}$	-0.701**	$-1.782^{***}$	-32.02***	-0.325***
	$U_2$	$4.956^{***}$	$-0.775^{***}$	$-2.521^{***}$	-48.85***	$-0.456^{***}$
	$U_1$	$4.400^{***}$	-1.012***	$-3.509^{***}$	$-68.35^{***}$	$-0.554^{***}$
	$U_0$	$3.156^{***}$	$-1.154^{***}$	-5.393***	$-97.44^{***}$	-0.920***
	$L_3$	$-1.751^{***}$	-0.143	$-1.645^{***}$	-29.96***	-0.248***
	$L_2$	$-1.526^{***}$	-0.923***	-2.209***	$-43.17^{***}$	-0.434***
	$L_1$	$-0.751^{**}$	$-1.097^{***}$	-3.809***	$-71.70^{***}$	-0.638***
	$L_0$	$1.129^{***}$	$-1.794^{***}$	-4.935***	-98.16***	-0.772***
Low	$U_3$	6.783***	-0.010	-4.009***	-88.84***	-0.017
	$U_2$	$6.014^{***}$	$0.475^{*}$	$-5.420^{***}$	$-120.1^{***}$	-0.097*
	$U_1$	4.898***	0.117	$-7.106^{***}$	$-143.0^{***}$	-0.146***
	$U_0$	$3.800^{***}$	0.144	$-9.194^{***}$	$-186.4^{***}$	-0.396***
	$L_3$	$-2.101^{***}$	-0.387	$-2.861^{***}$	-79.67***	-0.045
	$L_2$	$-1.578^{***}$	-0.344	-5.108***	$-129.0^{***}$	-0.060
	$L_1$	-0.511	-0.456*	$-6.182^{***}$	$-143.9^{***}$	-0.202***
	$L_0$	0.832	-0.761*	-8.718***	-200.3***	-0.439***
			BI	МЕ		
High	$U_3$	6.353***	0.170	$-1.421^{**}$	-10.55	-1.108**
	$U_2$	$5.398^{***}$	-0.082	$-1.893^{***}$	$-22.17^{***}$	-1.470***
	$U_1$	4.292***	-0.285	-3.035***	$-31.42^{***}$	$-1.642^{***}$
	$U_0$	1.873**	$-0.921^{***}$	-3.930***	-36.04***	$-2.957^{***}$
	$L_3$	$-3.425^{***}$	0.037	$-0.935^{*}$	$-13.29^{*}$	-0.576
	$L_2$	$-2.258^{**}$	0.033	$-1.291^{***}$	$-14.07^{**}$	-1.148***
	$L_1$	$-1.799^{*}$	-0.336*	$-2.599^{***}$	-29.43***	-0.992***
	$L_0$	0.916	-1.008***	-3.436***	$-34.17^{***}$	-2.390***
Low	$U_3$	7.474***	0.041	-3.601***	-20.65*	-1.005**
	$U_2$	$6.047^{***}$	-0.005	$-4.487^{***}$	-37.87***	-0.512
	$U_1$	4.756***	-0.200	$-5.547^{***}$	-53.80***	-0.775**
	$U_0$	$3.028^{**}$	-0.433***	$-6.442^{***}$	$-62.87^{***}$	-1.600***
	$L_3$	-3.437*	-0.293	-0.950	-14.22	-0.394
	$L_2$	-1.501	-0.312	$-2.542^{**}$	-33.98***	-0.500
	$L_1$	-0.698	$-0.524^{***}$	$-3.182^{***}$	-52.08***	-0.442
	$L_0$	1.957	-0.786***	-4.486***	-62.22***	-1.420***
Breuse	ch-Pagan	tests: The null	of independence i	s rejected at th	ie 1% level ( $p =$	0.000).

Table A.4: Trading behavior close to volatility interruptions subdivided by high and low high-frequency trading activity based on the 15-minute observation period reported separately for each market.

Results Subdivided by Market and High and Low HFT Activity (5 Minutes) This table reports the results of the regression model described in Equation (2) for sub-samples with high and low high-frequency trading activity separately for each market. The dummy variables  $U_d$  $(L_d)$  indicate how far a 10-second interval is away from the price limit in case of an upper (lower) limit hit. The ranges of distance to the limit (in bps) are  $I_3 = [200,150)$ ,  $I_2 = [150,100)$ ,  $I_1 = [100,50)$ ,  $I_0 = [50,0]$ . Controls for liquidity, order imbalance, and cumulative trading volume are included. Note: \* p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001.

		Trading Aggressiveness		Trading Activity		Volatility					
HFT	Coef.	Return $(R_t)$	Trade Size $(S_t)$	Trades $(T_t)$	Volume $(V_t)$	Volatility $(SD_t)$					
Xetra											
High	$U_3$	$5.116^{***}$	-1.014*	$-1.508^{***}$	$-37.44^{***}$	-0.319***					
	$U_2$	$4.917^{***}$	-0.502	$-1.599^{***}$	-37.86***	-0.301***					
	$U_1$	4.018***	$-1.340^{***}$	$-2.402^{***}$	-54.08***	$-0.523^{***}$					
	$U_0$	$2.813^{***}$	$-2.245^{***}$	$-3.374^{***}$	-69.98***	$-0.851^{***}$					
	$L_3$	$-1.765^{***}$	-0.455	$-1.029^{***}$	$-27.49^{***}$	0.086					
	$L_2$	$-0.674^{**}$	-0.873**	$-1.602^{***}$	-37.73***	-0.322***					
	$L_1$	0.060	$-1.351^{***}$	$-2.210^{***}$	-50.02***	$-0.519^{***}$					
	$L_0$	$1.111^{***}$	-2.440***	$-3.197^{***}$	-69.45***	-0.796***					
Low	$U_3$	6.037***	-0.230	-2.495***	-57.31***	-0.116					
	$U_2$	$5.716^{***}$	0.635	-3.780***	-84.98***	0.023					
	$U_1$	$4.780^{***}$	0.203	$-4.515^{***}$	$-104.4^{***}$	-0.258***					
	$U_0$	$3.551^{***}$	-0.693*	$-5.831^{***}$	$-137.7^{***}$	-0.395***					
	$L_3$	$-1.162^{***}$	-0.573	$-1.679^{***}$	$-63.11^{***}$	0.074					
	$L_2$	-0.510	-0.243	$-2.667^{***}$	-77.57***	-0.016					
	$L_1$	0.340	-0.962***	-4.086***	$-107.5^{***}$	-0.146***					
	$L_0$	$1.403^{***}$	-1.313***	$-5.457^{***}$	-143.5***	-0.310***					
BME											
High	$U_3$	6.520***	0.569	-0.489	32.12	0.625					
	$U_2$	$3.965^{***}$	$0.764^{**}$	-0.441	11.09	-0.824**					
	$U_1$	$2.745^{***}$	-0.175	$-1.615^{***}$	10.81	$-0.911^{***}$					
	$U_0$	0.655	-0.810***	$-2.488^{***}$	4.196	-1.880***					
	$L_3$	$-3.695^{***}$	-0.050	-0.862**	-33.30	-0.610					
	$L_2$	$-2.636^{***}$	-0.090	$-0.954^{***}$	8.962	-0.482					
	$L_1$	-1.216	-0.297	$-1.320^{***}$	20.37	-0.905***					
	$L_0$	0.268	-0.989***	-2.297***	-3.097	$-1.771^{***}$					
Low	$U_3$	8.998***	-0.112	-2.126***	-16.77	-1.294**					
	$U_2$	7.356***	-0.032	$-2.371^{***}$	$-19.88^{**}$	$-0.917^{*}$					
	$U_1$	$6.243^{***}$	-0.249	$-3.341^{***}$	$-25.37^{***}$	-0.770**					
	$U_0$	$3.912^{***}$	-0.956***	$-4.655^{***}$	$-53.19^{***}$	$-1.539^{***}$					
	$L_3$	0.494	0.175	-0.441	-16.63	-0.053					
	$L_2$	1.675	-0.625	-1.751**	$-24.69^{**}$	$-0.724^{*}$					
	$L_1$	1.408	-0.017	$-1.073^{*}$	$-26.71^{***}$	-0.372					
	$L_0$	$2.974^{***}$	$-1.566^{***}$	-3.892***	-53.62***	-1.434***					
Breuse	Breusch-Pagan tests: The null of independence is rejected at the 1% level $(p = 0.000)$ .										

Table A.5: Trading behavior close to volatility interruptions subdivided by high and low high-frequency trading activity based on the five-minute observation period reported separately for each market.

disclosed and undisclosed price limits. The dummy variables $U_d$ ( $L_d$ ) indicate how far a ten-second											
interval is away from the price limit in case of an upper (lower) limit hit. The ranges of distance to the											
limit (in bps) are $I_3 = [200,150), I_2 = [150,100), I_1 = [100,50), I_0 = [50,0]$ . Controls for liquidity, order											
imbalance, and cumulative trading volume are included. Note: * $p < 0.05$ , ** $p < 0.01$ , *** $p < 0.001$ .											
		Trading Aggressiveness		Trading Activity		Volatility					
Price Limits	Coef.	Return $(R_t)$	Trade Size $(S_t)$	Trades $(T_t)$	Volume $(V_t)$	Volatility $(SD_t)$					
Disclosed	$U_3$	8.085***	0.115	$-1.460^{***}$	-8.388	-0.648*					
	$U_2$	$6.200^{***}$	0.313	$-1.690^{***}$	-10.82	-0.996***					
	$U_1$	$5.029^{***}$	-0.343*	$-2.734^{***}$	-14.70	$-0.955^{***}$					
	$U_0$	$2.850^{***}$	$-1.017^{***}$	$-3.770^{***}$	$-37.10^{***}$	$-1.892^{***}$					
	$L_3$	-1.028	-0.010	-0.860*	$-42.60^{**}$	-0.400					
	$L_2$	-0.032	$-0.419^{*}$	$-1.665^{***}$	-30.09*	$-0.755^{***}$					
	$L_1$	0.705	-0.325	$-1.679^{***}$	-18.10	$-0.904^{***}$					
	$L_0$	$2.242^{***}$	$-1.358^{***}$	-3.360***	$-46.31^{***}$	-1.845***					
Undisclosed	$U_3$	5.759***	-0.727*	-2.515***	-57.79***	-0.222***					
	$U_2$	$5.451^{***}$	-0.121	-3.333***	$-74.55^{***}$	-0.186***					
	$U_1$	$4.429^{***}$	-0.938***	$-4.112^{***}$	$-92.74^{***}$	$-0.454^{***}$					
	$U_0$	$3.124^{***}$	$-2.170^{***}$	-5.213***	$-115.20^{***}$	-0.726***					
	$L_3$	$-1.358^{***}$	$-0.627^{*}$	-1.800***	$-57.28^{***}$	0.045					
	$L_2$	$-0.510^{**}$	-0.663**	$-2.718^{***}$	$-71.38^{***}$	-0.181***					
	$L_1$	0.285	$-1.405^{***}$	-3.832***	$-92.94^{***}$	-0.375***					
	$L_0$	$1.343^{***}$	-2.360***	$-4.976^{***}$	$-118.10^{***}$	$-0.641^{***}$					
Breusch-Pagan tests: The null of independence is rejected at the 1% level $(p = 0.000)$ .											

Results Subdivided by Disclosed and Undisclosed Price Limits (5 Minutes) This table reports the results of the regression model described in Equation (2) for sub-samples with

Table A.6: Trading behavior close to volatility interruptions subdivided by disclosed and undisclosed price limits based on the five-minute observation period.